

Contents lists available at ScienceDirect

Applied Energy



journal homepage: www.elsevier.com/locate/apenergy

Time-varying cost modeling and maintenance strategy optimization of plateau wind turbines considering degradation states

Huakang Tang^a, Honglei Wang^{a,b,*}, Chengjiang Li^a

^a School of Management, Guizhou University, Guiyang 550025, China

^b Key Laboratory of "Internet+" Collaborative Intelligent Manufacturing, Guiyang 550025, China

HIGHLIGHTS

• A multi-objective optimization model for maintenance decision of plateau wind turbine.

• The time-varying properties of downtime, distribution costs are modeled.

• The constant cost assumption is changed by calculating the time for device degradation.

• The run time of each degraded state of the wind turbine is calculated.

• The maintenance intensity and frequency are determined according to the maintenance target.

ARTICLE INFO

Keywords: Operation and maintenance Plateau wind energy Complex system Maintenance decision Multi-objective optimization

ABSTRACT

Plateau wind power has great potential in reducing carbon emissions; however, compared with other renewable energy, its economics still need to be improved. As an effective approach to enhance its economic feasibility, maintenance strategy optimization aims to reduce maintenance costs per kilowatt-hour and extend equipment lifespan. This paper proposes a multi-objective optimization model for the maintenance decision-making of plateau wind turbines that considers the degradation state. It incorporates: i) modeling the maintenance process of plateau wind turbines by combining time-based and state-based methods; ii) considering the time-varying maintenance costs in complex environments; and iii) employing a multi-objective optimization method to find the optimal strategy that meets maintenance requirements. The complexity considered in the model mainly includes the randomness of the operating duration for each equipment state, the temporal variability of equipment distribution and installation costs, and the uncertainty in maintenance effectiveness. The proposed optimization method is applied to a wind farm in the Yunnan-Guizhou Plateau, China.

The results indicate that traditional maintenance strategies underestimate maintenance costs and equipment lifespan losses. Compared with conventional maintenance strategies, this method can reduce equipment maintenance costs by 24.07 % and extend its operating life by 11.58 %. Additionally, this paper has conducted a series of parametric analyses to enhance the generalization performance of the model. The proposed method effectively addresses the economic issues of plateau wind turbine maintenance and provides a valuable decision-making tool for guiding the long-term maintenance of wind turbines in complex environments.

1. Introduction

In recent years, the global decarbonization trend has led to a rapid expansion of the share of renewable energy [1]. As nearly all suitable land for wind energy development in flatlands has been allocated, the focus of wind power development is shifting towards more complex environments such as oceans and plateaus [2–4]. Except for the

Antarctic continent, plateaus account for 30 % of the total land area [5]. The plateau region has a fragile environment and outdated power facilities, and residents rely on burning biomass for energy for a long time, posing significant challenges to the local ecosystem [6]. However, the plateau contains abundant wind resources [7], offering hope to overcome this tremendous challenge and improve the living conditions of residents.

The availability of flatland wind power has reached a satisfactory

* Corresponding author at: School of Management, Guizhou University, Guiyang 550025, China. *E-mail address:* hlwang@gzu.edu.cn (H. Wang).

https://doi.org/10.1016/j.apenergy.2024.124464

Received 19 May 2024; Received in revised form 7 August 2024; Accepted 6 September 2024 Available online 12 September 2024 0306-2619/© 2024 Elsevier Ltd. All rights are reserved, including those for text and data mining, AI training, and similar technologies.

Nomenclature		c_k	Maintenance resource ratio
		η	Maintenance effect influence factor
O&M	Operation and Maintenance	ε_k	Maintenance effect
PSO	Particle Swarm Optimization	v_k	Rate of degradation
NSGA-II	Non-dominated Sorting Genetic Algorithm II	cdf_1	Cumulative distribution function of Gamma process
RUL	Remaining Useful Life	ρ	Preventive maintenance cost coefficient
CM	Corrective Maintenance	N_k	The number of times thekth state is maintained
PM	Preventive Maintenance	γ _k	Lifetime decay rate
FFMS	Fixed Frequency Maintenance Strategy	f_k	The maintenance frequency of thekth state (times/year)
FRRS	Fixed Resource Ratio Strategy	cdf_2	Cumulative distribution function of Exponential
MOOS	Multi-objective Optimization Strategy	- 2	distribution
HFMS	High Frequency Maintenance Strategy	t_k	The running time of thekth state (month)
LFMS	Low Frequency Maintenance Strategy	\overline{C}_{C}	Average transportation and installation costs (\$)
Symbols		\overline{C}_S	Average downtime cost (\$)
a.	Distribution to installation cost rate	\overline{C}_I	Regular maintenance cost (\$)
01	Corrective maintenance cost negality factor	C_{F}	Equipment acquisition cost (\$)
ψ_1	Confective maintenance cost penalty factor	T_{s}^{-}	Down time (month)
θ_2	Downtime cost rate	0	

level (about 95 % to 97 %) [8]. If wind turbines (WTs) of the same type are installed on the plateau, their availability will significantly decrease, leading to a reduction in lifespan and an increase in maintenance costs. India, Denmark, China, and other countries have greatly tried to practice plateau wind energy development [9–11]. However, due to environmental complexities such as wind shear [12], turbulence [13], low temperature [14], the challenges of developing wind power on plateaus are immense. Operating and managing WTs is complex, and their lifespan will be significantly affected [15]. The cost of plateau wind power is mainly composed of construction costs and operation and maintenance (O&M) costs [16]. For wind farms that have been put into operation, improving the efficiency of operation and maintenance, and extending the service life of equipment are very important for enhancing their economic performance.

High operation and maintenance costs, coupled with limited service life of equipment, are among the main constraints facing the development of wind energy on the plateau [17]. The optimization of maintenance strategies [18] aims to provide maintenance decision-making basis for wind farm operators to enhance the economic operation capacity of WTs [19]. Maintenance strategy optimization aims to provide maintenance decisions for wind farm operators to improve the economic operation capability of WTs. A wind farm with a capacity of 100 MW has a fixed asset investment of 150 million US dollars [20], and the annual depreciation cost is very high. Optimizing the maintenance strategy could lead to cost savings of 30-40 %. Compared with a single goal [21-23], decision-makers often focus on multi-dimensional maintenance effects, such as minimizing maintenance costs [24] and maximizing economic benefits [25]. Due to the environmental complexity of the plateau environment, it is often difficult to find an optimal multiobjective maintenance strategy. For example, low temperatures lead to icing of roads and blades due to significant seasonal temperature differences in the plateau [26], which incurs high component transportation and installation costs [27]. In contrast, such time-varying costs are typically not a concern in plain regions.

Existing research mainly focuses on state-based [28] and time-based [29] maintenance decisions. The time-based strategy assumes that the equipment's lifespan follows distributions such as the Weibull or Gamma distribution, enabling the scheduling of periodic maintenance to improve equipment reliability [30]. Based on the "constant cost assumption," time-based maintenance strategies have yielded many useful conclusions [31]. However, as the environment faced by WTs becomes increasingly harsh, the applicability of this assumption has significantly decreased. Due to the progress of sensor technology and artificial intelligence technology, the state of WT components can be

monitored quickly [32]. State-based strategies plan the resources required for equipment maintenance in advance according to real-time status [33], but this approach gives less consideration to time information. In the complex plateau environment, WTs require not only periodic maintenance schedules but also plans for the allocation of equipment when immediate failures occur [34]. This necessitates that operators conduct a comprehensive analysis that considers the degradation characteristics of the equipment and its status information. Addressing the issue of not considering time information in state-based methods during the process of breaking the "constant cost assumption" is crucial for plateau wind power maintenance modeling. Therefore, the maintenance strategy considering time and state information can more accurately describe the situation in the equipment O&M process, and it is necessary to consider the time-varying maintenance cost caused by a complex environment.

This paper comprehensively employs state-based and time-based periodic maintenance strategies to address the production losses and lifespan reduction of WTs caused by the complex environment of plateaus. It utilizes stochastic processes to describe equipment degradation characteristics and establishes a time-varying cost maintenance decision model. By optimizing resource ratios and maintenance frequencies during the maintenance process, the aim is to lower the maintenance cost and longer the operational lifespan.

In summary, the contributions of the paper are:

- 1. In response to the complexity of the plateau environment, this paper employs multiple interrelated objectives to represent the economic aspects of operation and maintenance for WTs.
- 2. Given the single angle of traditional maintenance modeling, this paper combines the state-based and time-based maintenance models to describe plateau WTs' degradation and maintenance process more accurately.
- 3. This paper models the time that equipment degradation states have experienced as a stochastic process, providing data for the timevarying nature of costs, thereby solving the problem that the "constant cost assumption" is not applicable to the study of the economic feasibility of wind power in complex environments.

The remainder of this paper is organized as follows: Section 2 identifies the contributions of the paper through a literature review and proposes the research question. Section 3 presents the methodology used to address the research question. Section 4 discusses the relevant parameters of the maintenance optimization decision model. Section 5 includes the analysis and discussion of results, using a wind farm in the Yunnan-Guizhou Plateau of China for case analysis. Finally, Section 6 provides a summary and conclusion of this paper.

2. Literature review

2.1. Maintenance cost of plateau WTs

Almost all suitable land for wind energy development in flatlands has been incorporated into planning in recent years. In contrast, wind farms, as an infrastructure providing clean energy, must be expanded worldwide [35,36]. Plateaus account for 40 % of the world's land area, and the plateau wind farm has begun to provide electricity to the Qinghai-Tibet Plateau. Wind farms in the Pamir Plateau [37], Yunnan-Guizhou Plateau [38], and Pyrenees Mountains [39] have been operating for decades. In plateau, environmental factors such as low wind density, wind shear, and extreme temperatures [15] affect the O&M of WT units, resulting in higher O&M costs compared to flatlands. In terms of power supply, plateau WT have also encountered issues such as high downtime rates, unstable power supply, and low economic benefits. These problems have hindered the development of wind energy in plateau.

Due to the significant temperature difference between winter and summer, as well as the frequent freezing of roads and WT blades in winter, the maintenance cost of plateau wind power fluctuates considerably throughout the year [40]. Gyatso et al. [5] studied the operation performance of WTs on the Qinghai-Tibet Plateau but only considered maintenance decisions from the perspective of equipment selection. The load of wind power equipment changes with the wind conditions and market demand. Due to the spatial heterogeneity of wind power distribution [41], the load of the same type of WT in a wind farm may vary significantly throughout a year. However, most maintenance studies of WTs at sea or on the flatland rarely consider the characteristics of the environment and the equipment itself [25,42,43]. This paper integrates degradation states, runtime, and environmental factors such as freezing and wind speed fluctuations to establish a maintenance decision model for plateau wind power generation equipment. This aims to address the issue of current maintenance practices for plateau WTs neglecting environmental complexity.

2.2. Maintenance strategy

When wind power equipment begins to degrade, the decision regarding maintenance resources and maintenance frequency required can effectively reduce the cost of power generation [44]. In the early stage of equipment degradation, abandoning or directly replacing it will lead to higher costs. On-site maintenance can only formulate optimal maintenance strategies based on the degradation condition of each equipment, thereby reducing maintenance costs, and extending the equipment's lifespans. Many studies regard maintenance as a method that can improve the degree of degradation. Still, the evolution of the degradation state of most mechanical equipment is irreversible, such as cracks and defects will not heal automatically [45], such as WT gears [46], bearings [47], etc., so the operation life can only be prolonged by slowing down the degradation speed.

Time-based maintenance typically utilizes stochastic processes to describe the age of equipment [31], and maintenance is conducted based on the age and operational cycle of the equipment. Time-based maintenance strategies typically include the "constant cost assumption", which assumes that maintenance costs are fixed and unchanging, or that maintenance can restore the equipment to as good as new. Under the" constant cost assumption" [48], time-based maintenance decision optimization methods have yielded many interesting conclusions, such as the existence of an optimal maintenance interval within a finite horizon that minimizes the maintenance cost of the equipment [49]. However, changes in the environment have assumed of the "constant cost assumption" no longer applicable to real-world situations. For example, the change of seasons leads to different maintenance costs each

month, and market prices cause fluctuations in the prices of WT spare parts, etc. Within a plateau wind farm, even if two WTs are of the same age, their degradation levels may differ due to different loads and operating conditions [50]. Regardless of the equipment's age, its degradation state is intuitive information that reflects the equipment's current condition, such as a high risk of teeth breakage caused by the fracture of the gear root. State-based maintenance [28] directly uses condition information to make maintenance decisions on equipment. Equipment degradation depends on the amount of maintenance resources invested [51], and investing appropriate maintenance resources in each state is beneficial to prolong the lifespan of equipment [52]. State-based maintenance typically does not consider maintenance cycles and performs maintenance on the equipment immediately when a failure occurs. However, for heavy equipment like WT units, a large amount of resource allocation and maintenance personnel preparation is required before scheduled maintenance can be carried out [22,53]. Maintenance that considers both time cycles and status information can accommodate the development of long-term maintenance plans for wind farms and determine response measures for immediate failures.

During the operation of plateau WTs, the inevitable cost incurred by equipment degradation failure can be mitigated by reasonable maintenance measures, extending the equipment's lifespan. However, relying solely on state-based or time-based maintenance strategies is no longer sufficient to meet the maintenance needs of wind power in complex environments. With the rapid development of artificial intelligence and detection technology [54], sensor signals from industrial sites can be transmitted promptly and rapidly. Many studies have achieved online real-time equipment condition diagnosis with artificial intelligence algorithms. This paper assumes that the operating time of each state follows a stochastic process and combines state-based and time-based strategies to study wind power maintenance issues in complex plateau environments.

2.3. Time-varying costs

In the research on equipment maintenance modeling, Barlow et al. [31] introduced an age replacement strategy, which means that PM is carried out if the equipment reaches a critical age. This result proved to be the best choice under constant cost rate and thus extended to many studies such as integrated maintenance [55], production [56], and scheduling [57]. Most later studies [58,59] focus on extending the applicable conditions and less on the variable cost rate problem. For example, Papadopoulos et al. [60], based on environmental changes and changes in maintenance time, windows can perform operations when maintenance opportunities arrive to reduce production losses caused by equipment downtime.

With the change in equipment maintenance conditions, the conclusion under constant cost rate no longer applies to the maintenance of WT under variable cost rates in complex environments. Schouten et al. [40] considered the time-varying maintenance cost of a single WT. Still, they ignored the real-time state of the equipment and only described the WT degradation as a component following a stochastic process. In the operating environment of WTs, wind speed and direction change very fast, and the operation state of each WT is complex and changeable due to the equipment's operation temperature, humidity, and power generation plan [61]. Compared with the constant cost rate, the timevarying cost [62] can more accurately describe the changes in the plateau wind power O&M environment.

3. System description and maintenance strategy

3.1. General assumptions

To make the assumptions more consistent with the operating characteristics of the plateau WT and, at the same time, reduce the model complexity, we consider a system that can be simplified by a single component, such as considering only the most essential elements. Assuming that the system's state at time *t* can be represented by an observable random variable X_t , for example, the evolution of a gear crack, the process(X_t)_{t>0} is an increasing stochastic process.

In addition, some assumptions are as follows:

- At the initial time, the device's state isx₀. The device is in a brandnew state, and each CM replaces the previous component with an intact one.
- (2) The state detection is perfect and does not occupy the running time; that is, it can accurately and quickly detect the working state of the equipment. The degradation state is self-alarming; that is, the sensor detects the degradation state of the equipment all the time.
- (3) The failure of equipment can have significant consequences, and CM should be performed immediately upon occurrence of such failure.

3.2. Degradation modeling

During their operation, plateau WT units are often affected by freezing, squalls, and hail. These influencing factors are not continuously applied to the WTs but gradually cause damage to them over time.

The gamma process, which has been widely used to describe the degradation process of various industrial systems, is strictly monotonically increasing, which is in line with the degradation behavior of most devices. In addition, the gamma process [63] is the accumulation of countless small shocks, that is, discontinuity, and this characteristic is well in line with the degradation process of the mechanical structure of the plateau WT [64]. We assume that the degraded state between the *k*th state and the *k* + 1th state follows a gamma stochastic process(X_t)_{$t \ge 0$}, which has the following characteristics:

(1). $X_0 = 0, (X_t)_{t>0}$ is monotonically increasing.

(2). The independent increment $\tilde{X}_{k+1} - \tilde{X}_k$ of the degradation state between the *k*th state and the *k* + 1th state follows a gamma probability density function with shape parameter α_k and scale parameter β , as shown in Eq. (1).

$$f_{a_k,\beta}(\mathbf{x}) = \frac{1}{\Gamma(a_k)} \beta^{a_k} \mathbf{x}^{a_k-1} exp(-\beta \mathbf{x}) I_{\{\mathbf{x} \ge 0\}}$$
(1)

here, $\Gamma(\alpha_k) = \int_0^\infty \omega^{\alpha_k - 1} exp(-\omega) d\omega, \alpha_k, \beta > 0, I_{\{x \ge 0\}}$ is an indicative variable, $\alpha_k = \nu_k / \beta$, and ν_k is the rate of degradation between statek and statek + 1 at statex_k. The degree of PM can determine ν_k , that is, there is a functional relationship between the degree of PM and the rate of degradation. This paper argues that the degree of PM can change the average degradation rate; that is, PM with more resources can improve the average degradation degree, and cheap PM will even increase the degradation rate of equipment.

3.3. Maintenance strategy

Mechanical equipment will experience a variety of defect states during its life cycle. If the defect is repaired immediately, it will incur high costs. Advanced fault diagnosis methods often classify equipment degradation states into limited types [65], and corresponding maintenance decisions to limited kinds of degradation states can reduce the difficulty of maintenance decisions [66]. Fig. 1 illustrates the degradation paths of a WT gearbox.

The corresponding maintenance strategy is as follows:

In this study, finite degradation states are considered, and the degradation states are defined according to the actual industrial problems. Let the degradation state $x_k \in \{x_1, x_2, x_3, \dots, x_M\}$, equipment degradation develops in subscript order, such as gear crack change.



Fig. 1. Case of gearbox degradation path.

- 1. The climatic conditions and local load on the plateau are analyzed. It is found that when the wind speed is too high or the demand is low, the WTs cease operation.
- When *x_k* < *x*₁, the equipment is in a healthy state and does not need too much maintenance. It only needs to be inspected according to the daily work content.
- 3. When $x_m \le x_k \le x_n, m < n, m \ge 1$, the equipment is in a defective state, and PM with frequency f_k and maintenance intensity c_k is required according to the corresponding degradation state. The routine inspection currently is included in the PM process.
- 4. Whenx_k ≥ x_N, the equipment is in a failure state, and CM needs to be performed immediately, otherwise the failure will cause permanent damage to the whole system.

Fig. 2 is the decision-making flowchart for this paper, which is mainly divided into two parts based on the results of the status check. If the equipment is in a healthy state, perform routine maintenance. If the equipment has entered a degradation phase, use the methods of this paper to make decisions on the resources and maintenance frequency required for the maintenance process. Then execute this maintenance strategy until the equipment fails.

4. Model formulation

This section describes the formulation of the proposed model, which was developed from the research of Phuc Do et al. [67] and Schouten et al. [40]. The purpose of the maintenance model is to minimize the maintenance cost per kWh and maximize the operation time of the equipment. The decision variables are the maintenance frequency and resource input ratio at each stage of the WT maintenance. The maintenance cost depends on the stochastic preventive and time-varying CM costs. CM costs, including maintenance resource deployment, lifting equipment, and labor costs; PM costs, including replacement parts, labor costs, downtime costs, and downtime costs refer to the loss of revenue caused by WT replacement equipment.

4.1. Time-varying cost rate

As the road of the plateau wind farm is often relatively poor, it is prone to freezing in winter [68], which is an enormous challenge for the transportation of maintenance resources and causes varying resource distribution costs over time. In addition, the downtime cost needs to consider the plateau climate change, which has similarities with the resource distribution cost, and the downtime cost rate can be obtained according to the seasonal patterns of the wind field.

This paper determines the CM cost rate function and the downtime cost rate function. According to the relationship between the risk of extreme weather and time, considering the relatively high cost caused



Fig. 2. Decision-Making Flowchart.

by distribution and installation in the CM process, the cost rate function of distribution and installation is given by Eq. (2).

$$\varphi_1, t \in PeriodofFreezing$$

$$1, other$$
(2)

The CM cost multiplier during the freezing period is set to φ_1 , $\varphi_1 > 1$, while the resource distribution cost rate is 1 at other times. Winter is a high-incidence season for road icing, and the cost of maintaining resource distribution and installation is very high. As a penalty, we set a very high delivery and installation cost rate during the freezing period to avoid major component replacements during this time.

The characteristics of wind speed change in plateau areas may differ from those in flatlands. Firstly, the wind speed in the plateau area is generally lower compared to coastal regions. Secondly, the peak-tovalley variation trend of wind speed also differs from that in flatlands. The peak-to-valley variation of wind speed changes in the plateau area is affected by terrain, altitude, longitude, and latitude. Accurate fitting of the variation patterns can be achieved by utilizing local wind measurement station data.

According to the study of Schouten et al. [40], the wind change can generally be fitted as the following function, and the downtime cost rate can be obtained by simply fitting the parameters based on field data.

The change rate of downtime cost is given by Eq. (3),

$$\theta_2 = Asin(\omega t + \varphi_2) + B \tag{3}$$

where *t* is the downtime, *A*, *B*, and φ_2 are used as correction parameters, which are determined by the wind speed variation rule at the study site. According to the equipment degradation cycle length and inspection law, using a month as the maintenance time unit is more appropriate.

4.2. Preventive maintenance cost

When the equipment is in a particular state, the decision variables are the ratio of resources invested in each maintenance and the maintenance frequency of this state. The ratio of resources invested in each maintenance determines the degradation speed, where a higher resource allocation ratio leads to a reduced degradation rate. The maintenance frequency determines the state's running time, with an appropriate maintenance frequency prolonging the continuation time of the state, while a lower maintenance frequency reduces the running time [64].

4.2.1. Quality of maintenance

The cost of PM is usually determined by the level of resources allocated to the operation. The cost of maintenance increases with the increase of resources, and the quality of maintenance also improves. The degradation rate depends on the degree of maintenance, and the maintenance behavior may only sometimes improve the system. As evidenced by numerous cases in practice indicate that low-quality maintenance may accelerate the deterioration of equipment [69,70].

The maintenance quality depends on the resource $ratioc_k$ invested in each maintenance in the state x_k stage. According to the actual application, it can be set to the maintenance tool level, worker level and other factors that affect the maintenance effect.

When the equipment is in statek, the impact of each maintenance input resource ratio c_k on maintenance quality can be described by variable ε_k , which satisfies Eq. (4),

$$\varepsilon_k = 2(c_k)^\eta - 1 \tag{4}$$

here, c_k is the ratio of resources invested in each maintenance at the state $x \in (x_k, x_{k+1})$, whose value ranges from (0, 1). The value ε_k is in the range (-1, 1), indicating whether the maintenance was of low or high quality. η is the correlation factor, which describes the nonlinear relationship between maintenance resource ratio and maintenance quality and takes a nonnegative real number.

If the maintenance action of state k is CM, then the degradation rate after maintenance is reset $tov_0 = \alpha_0/\beta$. Here, α_0 refers to the shape parameter of the gamma process when the WT is in a healthy state, and β is the scale parameter of the process. If the maintenance is PM, the degradation rate after maintenance is Eq. (5).

$$\nu_k = \nu_{k-1}(1 + \varepsilon_k) \tag{5}$$

Eq. (5) represents the impact of maintenance quality on the degradation rate, and in the subsequent description, the degradation rate is used to model the maintenance cost, the slower the degradation after maintenance, the higher the cost of maintenance. Based on the degradation rate, the cost of PM can be evaluated as a function of the degradation rate. According to Eq. (5), we can obtain the recursive formula for the shape parameter α_k when the WT is in state x_k , which is $\alpha_k = \beta \nu_k$.

We assume that the degradation state between the *k*th state and the *k* + 1th state follows a gamma stochastic process $(X_t)_{t\geq 0}$, $cdf_1(x)$ is the cumulative distribution function of the degradation state *x*, and the probability of the device being in state $x \in (x_k, x_{k+1})$ is given in Eq. (6).

$$P_k(\mathbf{x}) = P\{\mathbf{x}_k < \mathbf{x} < \mathbf{x}_{k+1}\} = cdf_1(\mathbf{x}) - cdf_1(\mathbf{x}_k), (k = 1, 2, \cdots)$$
(6)

The cost of the PM phase is C_p ,

$$C_P = N_k \sum_{k=1}^{M} \rho c_k P_k(\mathbf{x}) \tag{7}$$

 ρ is the cost coefficient of the PM phase, which is determined by the actual situation of the wind farm. N_k is the maintenance number of the *k*th state, $N_k = t_k f_k f_k$ is the maintenance frequency at statek, and *M* is the number of degraded states that are divided.

PM strategy optimization can be applied to decision-making processes such as long-term maintenance resource procurement and optimal staff scheduling to determine the optimal maintenance strategy and resource allocation strategy.

4.2.2. Frequency of maintenance

To simulate the effect of maintenance frequency on the operation time of each state of the equipment, this paper assumes that the operation time of each state is determined only by the maintenance frequency of that state. This can be found in many practical studies [71]. For example, adding lubricating oil to the bearing at an appropriate frequency can keep the pitting area of the bearing rolling element unchanged. In contrast, the pitting area of the rolling component will become more prominent when the frequency is low, and the operation time of the equipment under the pitting defect state will be shortened.

When the equipment statex $\in (x_k, x_{k+1})$, the impact of the maintenance frequency f_k at stage k on the reduction in the running time of the equipment at this state can be expressed by a nonnegative continuous random variabley, which follows the exponential distribution with probability density function as in Eq. (8).

$$h(\mathbf{y}) = \gamma_k exp(-\gamma_k \mathbf{y}) I_{\{c_k \ge 0\}}$$
(8)

 f_k is the maintenance frequency of the device at phasek, and a different maintenance frequency is used in each state. Since the maintenance frequency is constant in a state, it is assumed that the equipment degradation rate is constant in this state, and the reduction in the running time of this state relative to the previous state due to degradation is determined by γ_k , γ_k as in Eq. (9).

$$\gamma_k = \log_a(f_k) \tag{9}$$

a is used as the basic unit of maintenance time, meaning that the year a is divided into equal parts. Due to the limitation of maintenance resources, wind farms often use a fixed duration as the basic maintenance unit to ensure wind power benefits. When a = 12, then the basic unit of maintenance time is one month, and if $f_k = 2$ currently, it means that maintenance is performed twice a year $f_k \in \mathbb{Z}^+$. Here, *a* must be greater than 1 to ensure that the equipment gets more than one maintenance opportunity per year.

At a certain frequency f_k , the probability of the reduction of the running time of a state x_k relative to the previous state x_{k-1} is given by Eq. (10).

$$\vartheta_k = cdf_2(\mathbf{x}_k) - cdf_2(\mathbf{x}_{k-1}) \tag{10}$$

 $cdf_2(x_k)$ denotes the cumulative probability function of $x \in (x_k, x_{k+1})$. The duration of the device in state k - 1 is t_{k-1} . After the maintenance of frequency f_k , the duration t_k in state k is given by Eq. (11).

$$t_k = t_{k-1}(1 - \vartheta_k) \tag{11}$$

That is, when the equipment is in a certain $statex_k$, the higher the maintenance frequency of this phase, the longer the duration of this phase. Since the device degradation is more severe than the previous state, we assume that the current state will not run longer than the previous state.

4.3. Corrective maintenance costs

Maintenance operations require corresponding resource allocation, and the seasonal turnover in the plateau region causes varying maintenance cost rates over time. When a fail-down event occurs, CM is necessary.CM requires deploying maintenance resources from external sources and assigning workers, which involves the remote movement of heavy equipment, which is highly costly due to the extreme temperatures in the plateau. Therefore, regarding equipment replacement costs, it is crucial to not only account for the equipment purchase cost but also factor in the changing equipment transportation cost [72]. Because it takes more time for maintenance resources and personnel to arrive at the wind farm in winter, and the wind in winter is not conducive to maintenance operations, CM will cause extended downtime and incur huge downtime costs.

The equipment replacement cost of plateau wind power includes the equipment purchase cost and the maintenance equipment assembly cost. The rise and fall of equipment acquisition cost is slight, and the acquisition cost of equipment in the whole life cycle is assumed to be a constant value C_E .

The change of maintenance equipment assembly cost rate with time is expressed by Eq. (12).

$$C_{C}(t) = \overline{C}_{C}\theta_{1}(t) \tag{12}$$

 \overline{C}_{c} is the average assembly cost generated by the primary equipment replacement of the important components of the plateau WT, and $\theta_{1}(t)$ is the resource distribution cost rate of the plateau wind farm. tis the time that the equipment needs to be replaced, depending on the total time of operation of the equipment and the month in which the equipment was just put into operation. Generally, when the equipment fails, the purchase demand is sent to the manufacturer immediately, and the manufacturer can ship the equipment to the wind farm in time.

The downtime cost mainly focuses on the loss of power generation due to wind loss during the downtime and is related to the specific time when the downtime occurs. The downtime cost during the equipment replacement period is given by Eq. (13).

$$C_{\mathcal{S}}(t) = T_{\mathcal{S}}\overline{C}_{\mathcal{S}}\theta_2(t) \tag{13}$$

 T_s is the average downtime duration and \overline{C}_s is the average downtime cost per unit time. *t* is the moment when the device is replaced.

Regardless of downtime or CM costs, operators do not want downtime when extreme cold occurs. Through the previous analysis, we can decide the degree of PM each time, which will change the degradation state experience time, which makes it possible to change the WT downtime. Specifically, we need to control the maintenance degree in the PM process and try to control the downtime at other times except the freezing period.

4.4. Regular inspection of costs

General industrial equipment will conduct regular inspections, whether in a defective or normal state. Although the regular inspection cycle is relatively long, it will still produce non-negligible costs. The difference between conventional and PM costs is that conventional maintenance considers the cost of transportation and personnel arrangement in the process of regular inspection. In contrast, PM cost is the cost of maintenance resources used in the maintenance process because PM requires maintenance resources that are more expensive than conventional maintenance.

Each inspection looks for the moment of lower wind speed in the future to reduce the impact of downtime. The inspection cost of the whole life cycle is given by Eq. (14).

$$C_I = \overline{C}_I N_I \tag{14}$$

Among them, N_I is related to the wind field environment and WT characteristics, $N_I = T/T_I$, and T_I is the routine inspection cycle. According to the maintenance regulations of the wind field, the unit needs to carry out routine maintenance of the WT within a fixed time. \overline{C}_I represents the average downtime, resource consumption, and labor cost incurred during routine maintenance.

4.5. Maintenance objectives

To evaluate the performance of the maintenance strategy, this paper takes minimizing the total maintenance cost of the life cycle and maximizing the operation time as the goal and finally obtains the maintenance frequency f_k and the resource investment ratio c_k when the optimal decision parameter state is k.

The total maintenance cost of the key equipment of the WT from operation to failure replacement is given by Eq. (15), which includes both preventive maintenance costs and corrective maintenance costs.

$$C_{all} = C_P + C_C + C_S + C_I + C_E$$
(15)

H is the running time of the equipment, as shown in Eq. (16).

$$H = \sum_{k=1}^{M} t_k \tag{16}$$

The maintenance requirement of the wind farm is to minimize the maintenance cost per kWh and maximize the operation time. The optimization objectives and constraints are as follows,

$$\min C = \frac{C_{all}}{H \cdot E} = \frac{\sum_{k=1}^{M} C_k P_k(x) + \overline{C}_C \theta_1(t) + T_S \overline{C}_S \theta_2(t) + \overline{C}_I N_I + C_E}{H \cdot E}$$
(17)

$$max H = \sum_{k=1}^{M} t_k \tag{18}$$

E is the amount of electricity generated per unit time, which can be obtained by consulting the wind farm operation log.

t = (H + initial time)//12in objectiveCmeans that the *initial time* passes through the total life length*H* and the specific month of shutdown (the remainder is taken to represent the month). *H*·*E* represents the total amount of electricity generated during the equipment's life cycle.

If the expected maintenance cost of the equipment at a certain stage exceeds the average CM cost, then CM is the more economical choice, so a constraint $\rho c_k N_k + t_k \overline{C}_S < \overline{C}_C + T_S C_S$ needs to be introduced, That is, the sum of the PM cost and the subsequent generation revenue at a certain stage needs to be less than the average CM cost $\overline{C}_C + T_S C_S$, otherwise it directly enters the CM procedure.

Therefore, the constraints are as follows,

$$\begin{cases} f_k \in \mathbb{Z}^+ \\ 0 < c_k \le 1 \\ \rho c_k N_k + t_k \overline{C}_S < \overline{C}_C + T_S C_S \end{cases}$$
(19)

According to Eqs. (17)–(18), the model has two objectives, with decision variables being the maintenance frequency for each state and the maintenance resource ratio. For the multi-objective optimization model, this paper employs the NSGA-II algorithm for solving, while for the single-objective optimization problem in the comparative experiment, this paper utilizes the PSO algorithm for solving.

5. Numerical example

5.1. Scenario set-up

The wind farm studied in this paper is in western Guizhou Province, China, at an altitude of 2700 m. Since the wind farm has been in operation for a considerable period, it has experienced various mechanical and electrical failures in a typical plateau environment. Among them, the fault with the most enormous single economic loss is attributed to the gearbox. To achieve the above maintenance objectives, the WT of the wind farm is taken as the research object, and the field data from the wind farm are collected to conduct a case study.

A WT has many components, but the mechanical components determine its maintenance costs. In addition to the gearbox, other components, such as generators, bearings, etc., choose the WT's maintenance cost. The method of the text can also be used as a module to construct a maintenance decision model for multiple components. Because other essential components can also divide their degradation states by monitoring techniques, this paper's method of combining state and time is still valid. Therefore, the case study part of this paper only considers the WT component that causes the most significant single economic loss –the gearbox.

Fig. 3 shows the monthly average wind speed change curve of the wind farm studied in the past 10 years [73]. Observing the ten-year wind data obtained in this paper, it can be found that the wind speed of the Yunnan-Guizhou Plateau wind farm studied in this paper has a trend of sinusoidal variation. To model concisely while also describing the characteristics of the data, we have also chosen the sine function for modeling.

It can be seen from the figure that the average wind speed is the maximum in March and the minimum in September. Using the fitting method to fit the scatter plot, Eq. (20) can be obtained,

$$\theta_1(x) = 0.7 \sin\left(\frac{\pi}{6}x - \frac{\pi}{12}\right) + 2.99$$
 (20)

The wind farm icing cycle occurs from the middle of December to the middle of February, a total of 2 months, during which there are many snow and freezing days. Due to the long CM cycle of the equipment, it is entirely possible to fall within this span, so it is necessary to avoid CM in this period [74]. This paper uses the penalty factor method to prevent the WT replacement event from occurring in the freezing period. Fig. 4 shows the monthly average weekly freezing days in the wind farm area in the last ten years. (See Fig. 5.)

Due to the progress of prediction technology, the rotating machinery of the plateau WT has been able to predict its operation stage roughly, and the average downtime caused by failure has been shortened from 2 to 3 months to half a month [75]. Although the equipment replacement cycle is significantly reduced, if the replacement occurs in the freezing season, it is still necessary to perform road and tower blade deicing, which will bring considerable time and economic costs.

The following are some parameters from the wind farms under study, Table 1 shows the cost coefficient versus the moment of commissioning of the equipment, \overline{C}_C is equipment transportation and installation cost, \overline{C}_S is downtime cost, \overline{C}_I is routine.

maintenance cost, and C_E is equipment acquisition cost. The units of the above costs are thousands of dollars (k\$), T_S downtime time is taken as maintenance unit in months.

The initial time here means that the WT was installed and put into operation in November. According to the reference [67], the shape parameters and scale parameters are selected as Table 2.

Given the highly nonlinear nature of the objective function, averaging can mitigate the impact of random fluctuations in individual experiments on the outcomes, leading to a more balanced set of solutions. Consequently, this paper employs a method of averaging results from multiple experiments for each experimental setup.

5.2. Results of fixed maintenance frequency

This section optimizes the policy using a fixed maintenance frequency. The fixed maintenance frequency data were collected from the wind farm described in the previous.



Fig. 3. Fitted plot of monthly mean wind speed.





Fig. 4. Period of Freezing.





Fig. 5. Comparison of maintenance cost per kWh for fixed input resource ratio.

Table 1

Cost coefficient and initial time.

$\overline{C}_C(k\$)$	$T_S(months)$	$\overline{C}_{S}(k\$)$	\overline{C}_I (k\$)	C_E (k\$)	Initial time
20	0.5	45	1	200	November

Table 2

Model parameters.

1			
α_0	β	η	ρ
1	2	2	0.1

section, and the input parameters are provided in Section 5.1. The FFMS focuses on the maintenance resources required for the equipment in each state and can be considered as a state-based maintenance strategy.

The unit of the above strategy is the number of maintenances per year. In the current practice of wind farm maintenance, the influence of maintenance frequency on the unit is rarely considered. Therefore, this section summarizes the two maintenance frequencies shown in Table 3 according to the actual situation of the wind farm to verify the influence of frequency on the maintenance effect.

Because the maintenance frequency is fixed, the service age of WT components depends entirely on the maintenance frequency through the

Table 3									
Fixed maintena	Fixed maintenance frequency strategy.								
Church									

Strategy	x_1	x_2	x_3	x_4	<i>x</i> ₅
HFMS	6	7	8	9	12
LFMS	2	3	4	6	12

analysis model; an objective has been fixed here, so it is converted into a single objective optimization problem. Using the PSO algorithm [76] with the particle number of 100 and 200 iterations, when the low-frequency maintenance strategy is used in the maintenance of the WT components, the cost of kWh can be reduced to 4.687\$/MWh by optimizing the maintenance resource ratio, and the life of the equipment is 261.7 months. When using the high-frequency maintenance resource ratio to 3.352\$/MWh, which is 278 months of life. The above conclusions show that the strategy of higher maintenance frequency in the early degradation stage will not only increase the cost of electricity per unit but can also prolong the service life of the equipment.

5.3. Results for fixed resource ratios

Currently, most wind power airports do not consider the impact of the change in input resource ratio on equipment life and production costs when performing maintenance. To compare the effects of different maintenance resource ratios on the life and production costs of the WTs, this section uses different maintenance resource ratios to optimize the maintenance of the plateau WTs. FRRS mainly optimizes the variable of maintenance frequency, which is consistent with the time-based maintenance strategy, and therefore can be regarded as an optimization of the time-based maintenance strategy.

According to the maintenance resource input ratio of each state of the WT gear, five maintenance strategies are designed, as shown in Table 4. The number represents the input resource ratio, which is distinguished according to the specific equipment. For example, the quality of maintenance supplies, workers' technical proficiency, and the condition monitoring level can be considered when maintaining the gearbox. The maintenance of fixed maintenance resource ratio is a multi-objective optimization problem. Here, the NSGA-II algorithm [77] with 100 particles and 300 iterations is adopted to optimize the problem, and the results are as follows. The NSGA-II algorithm incorporates an elitist strategy and a method for calculating crowd distance, thereby enhancing the efficiency and quality of solutions. It possesses strong robustness and effectiveness when dealing with multi-objective problems.

Where the ordinate is the maintenance cost per kWh, the unit is \$/MWh. Combined with Table 3, we can see that during the healthy and slight wear of the equipment, maintenance with a low resource ratio can reduce the kWh cost of the equipment life cycle to a greater extent. Maintenance using a large resource ratio may not be worth the loss in the early stage of equipment operation. Combined with the conclusion of the previous section, in the early stage of equipment operation, operation, and maintenance personnel need to maintain the equipment a small number of times, and the maintenance frequency can be higher. Still, it can use low-maintenance resources. In the middle and late stages of equipment operation, increasing the maintenance frequency and improving the resources invested in maintenance is necessary.

5.4. Model validation

To verify the validity and credibility of the model, this section studies the maintenance strategy of the WT in the plateau wind farm when it is put into operation at different times. The model considers equipment

Table 4Five fixed maintenance resource ratio strategies.

Strategy	<i>x</i> 1	<i>x</i> ₂	x_3	<i>x</i> ₄	x 5
Strategy One	0.01	0.05	0.6	0.8	0.9
Strategy Two	0.1	0.15	0.6	0.8	0.9
Strategy Three	0.1	0.2	0.6	0.8	1.0
Strategy Four	0.2	0.4	0.6	0.8	1.0
Strategy Five	0.6	0.7	0.8	0.9	1.0

distribution and installation costs over time, and the primary input parameters have been given in Section 5.1. Table 1 shows the parameter settings used by the NSGA-II algorithm to obtain the optimal solution. The population size of the algorithm is 100 individuals, the iteration is 300 generations, and everyone's fitness value is evaluated by the given calculation formula for the running time and the calculation formula for the maintenance cost of per MWh.

Fig. 6 shows the obtained Pareto front, which demonstrates the nondominated solutions for the device's expected lifetime and the cost per MWh. (See Figs. 7 and 8.)

Due to the high cost of distribution and installation, the high risk of abnormal climate, the wind farm studied in this paper does not carry out the loading and unloading operation of significant components from December to February of the following year. So, this paper chooses March, June, September, and November as the initial months for the WT to be put into operation. The resulting Pareto fronts are shown in Fig. 6 (a), 6(b), 6(c), and 6(e). Analysis of Fig. 6 shows that different operational times will lead to different optimal maintenance strategies for WTs. Putting the WT into operation in March can make it run longer, but the maintenance cost will increase in the optimization strategy. The reasons for the jump are analyzed in the parameter analysis section. Commissioning in June may shorten the WT's life but will not significantly impact maintenance costs. The service life of the WTs put into operation in September is reduced even more, and the maintenance cost is not significantly affected. For the WTs put into operation in November, using the maintenance strategy developed in this paper may extend the WT's life, but it will increase the maintenance cost. Table 5 shows the four Pareto-optimal strategies with the operational time of March, June, September, and November.

This paper selects four typical installation moments for WT equipment to optimize the O&M strategy. Combined with Fig. 6, the WT maintenance must be carried out at a high frequency in the early stage of equipment degradation, requiring a small investment in maintenance resources, but this is very effective for WT gearbox maintenance.

Table 6 compares the maintenance strategies and their effects. Combining the results of Sections 5.2 and 5.3, it can be found that when using the FFMS, which is equivalent to using a state-based method, even if the optimal state-based strategy results in a maintenance cost of \$4.02/MWh, the multi-objective optimization method in this paper can still reduce the maintenance cost by 24.07 %. At the same time, by comparing the operating time, it can be found that in the settings of this paper, using the state-based maintenance strategy will result in a loss of 11.58 % of the lifespan. When using the FRRS, i.e., the time-based maintenance strategy, the optimal maintenance frequency results in an average lifespan of 289.3 months, while the multi-objective optimization method in this paper can increase the operating time by 3.69 %. This also means that using the time-based maintenance strategy will result in an economic loss of 18.21 %. After comparing with FFMS and



Fig. 6. Time-varying costs considering time and state maintain the optimization results.



Fig. 7. PM effects for different_η.

FRRS, the results show that the multi-objective maintenance strategy in this paper can make the plateau WT have a longer lifespan with lower maintenance costs.

Furthermore, the optimization results also indicate that the FRRS adopted by traditional wind facilities, which means that the maintenance resources in the early and later stages of equipment operation are the same, affecting later-stage performance of the equipment. It is necessary to invest in maintenance with lower resource ratios in the early stages of equipment operation and then transition higher ones in the last stages. Only with such a maintenance strategy can minimize the cost of electricity consumption during operation. Unlike previous findings, the maintenance party needs to employ a higher maintenance frequency in the early stage and maintain a higher maintenance frequency in the later stage to maximize the equipment's operational lifespan. Considering all factors, during the maintenance process of plateau WTs, low resource ratios but multiple maintenance are required in the early stage of equipment operation. While in the later stage of equipment operation, high-level resources and higher frequency maintenance are necessary to optimize kilowatt-hour maintenance cost and operational lifespan.

5.5. Applicability analysis

To demonstrate that the maintenance strategy considering both time and state information can accurately represent the actual operation and maintenance process of equipment under the complex operating environment of the plateau, this section simulates the maintenance process of WTs in different scenarios by changing model parameters. Since plateau wind power has a relatively poor economic performance compared to plain wind power, the main comparison object in this section is plain wind power.

The air density on the plateau is low, and when the wind force is small, it does not meet the starting conditions for WTs at all. Most plain wind farms do not have this problem, and their WTs can almost always meet the power generation conditions throughout the year. Therefore, in plain areas, no matter when equipment maintenance and replacement are carried out, there will be downtime costs. However, in plateau, if maintenance plans are executed when there is no wind, it is possible to avoid huge downtime costs. Therefore, this paper uses the annual



Fig. 8. Non-dominated solutions for different_η.

Table 5

Pareto strategies for different input running times.

-						
Time	Variable	<i>x</i> ₁	<i>x</i> ₂	<i>x</i> ₃	<i>X</i> ₄	<i>x</i> 5
March	c_k	<0.1	<0.1	0.1-0.63	0.1–0.54	0.8–1
	fk	6–9	6–12	6–10	8–10	9–12
	c_k	<0.1	<0.1	0.8–1	0.4–0.6	0.6 - 1
Julie	f_k	6–8	6–9	6–12	8-12	9–12
Contombor	c_k	<0.1	0.2-0.25	0.6-0.93	0.5-0.81	0.6-0.77
September	f_k	7–9	7–11	5–9	4–10	4–8
November	c_k	<0.1	<0.1	0.6-0.9	0.16-0.44	0.57-0.74
november	f_k	8–12	9–12	8-12	9–12	8–12

Table 6

Comparison of maintenance strategies.

Strategy	Maintenance cost(\$/MWh)	Run time(months)
FFMS	4.02	268.85
FRRS	3.7324	289.3
MOOS	3.0525	299.97

average wind speed to replace the downtime cost change rate function of the plateau environment, indicating that the plain area can meet the power generation conditions throughout the year. Since the phenomenon of road icing rarely occurs in plain, and even in winter, effective maintenance of wind power equipment can be carried out, the assembly cost rate is set to 1, indicating that there is no additional assembly cost generated by environmental changes.

Table 7 presents the optimal solutions obtained using this model under different scenarios. In the scenario settings, two scenarios are set for the plain: the first is where the wind is relatively stable and can meet the power generation conditions throughout the year; the second is where there are slight wind fluctuations within a year, but the fluctuations are small, and the assembly cost rate for both scenarios is set to 1. According to surveys, on the Qinghai-Tibet Plateau, the wind fluctuations are relatively small throughout the year, but the freezing period is longer, resulting in higher equipment assembly costs; while on the East African Plateau, there are larger wind fluctuations throughout the year, but the freezing time is shorter; on the Yunnan-Guizhou Plateau, there are larger wind fluctuations throughout the year, and the time of freezing is also longer. Implementing the maintenance strategy proposed in this paper under extreme conditions can demonstrate the applicability of the method presented.

From the table, the plain with stable wind power can obtain the highest benefit, including the lowest maintenance cost per kWh and the longest lifespan. As the environment deteriorates, the cost per kWh gradually increases, and the lifespan of the equipment also gradually shortens. In the plateau scenarios, the data from the extreme environment comes from a wind farm in the western part of Guizhou Province, where the monthly average wind speed fluctuations are significant, and the freezing period is also relatively long. At the same time, we designed maintenance experiments for several other wind farms, studying the impact of different environmental factors on the maintenance effect of plateau wind farms by controlling variables. The experimental results

Table 7

Applicabil	Applicability Analysis under Different Scenarios.					
Scenarios	5	Maintenance cost (\$/MWh)	Run time (months)			
	Wind stability	2.8182	302.17			
Plain	Low wind fluctuations	2.8324	301.46			
	The wind is choppy	3.0350	300.86			
Plateau	Long freezing period	3.0436	300.37			
	Extreme environment	3.0525	299.97			

show that during the execution of the maintenance strategy provided in this paper, compared to scenarios with larger wind speed fluctuations, scenarios with longer freezing periods will have a greater maintenance cost per kWh, and the lifespan of WTs will be shortened. This indicates that freezing periods has a greater impact on the economic performance of plateau wind power, while the impact of monthly average wind speed fluctuations is slightly lower.

5.6. Sensitivity analysis

The parameters and data analyzed above are from a plateau wind farm in the west of Guizhou, China, and the selected parameters may deviate from other plateau wind farms. This section analyzes the crucial parameters in the model to analyze the generalization performance of the model.

5.6.1. Maintenance effect influence factory

 η is a parameter that influences the ratio of maintenance input resources on the quality of PM. Different operators have different operation levels. To simulate the maintenancelevels of varying wind power operators or maintainers, this paper uses Eq. (4) to describe this relationship, and its characteristics are shown in Fig. 7.

To study the change of η on the maintenance effect, this section sets up five control experiments with different values of η . Fig. 8 (d) is the nondominated solution obtained in the previous section, which is put into Fig. 8 for the convenience of comparison.

Here, the values of η are chosen as 0.1, 0.3, 1, 2, 2.5, and 5.

 η represents the impact of maintenance input resources on maintenance quality, which generally depends on the workability of the maintainer. Wind farms with higher performance can convert maintenance resources into corresponding maintenance quality, while wind farms with lower performance have difficulty converting invested maintenance resources into corresponding maintenance quality. Since the value of the input resource ratio is a decimal between 0 and 1, and the input resource ratio is very high in the later stage of maintenance, the larger η is, the faster the transformation efficiency changes, and the WT maintenance effect is more affected by the resource ratio, which leads to the jump of the solutions in Fig. 8 (e) and Fig. 8 (f). When *n* is between 0.1 and 2, it is more convenient to use this model for maintenance decision making, when $\eta > 2$, the non-dominated solution will produce some jumps, which is unfavorable for the decision process.

5.6.2. Sensitivity analysis of PM cost ratiop

This section studies the impact of varyingpon the maintenance effect. As the scaling factor of the cost of the PM phase in the maintenance cost of the whole equipment life cycle, the value of pcan be adjusted according to the actual situation of the wind farm. The wind power equipment selected in this paper does not consider the most extreme situation in the plateau environment, and the value of ρ of this equipment in the PM stage is 0.1. Due to the change in environment and the selection of model parameters, we consider the scaling factor $\rho \in (0, 0.25)$ to comprehensively show the change in the maintenance cost of wind power equipment in the complex plateau environment.

Table 8 shows the impact of the change in ρ on the two objectives.

When ρ is small, it means that the cost of the PM stage is relatively small, and the influence of the local environment on the degradation of the WT is not significant, while when ρ is large, it means that the impact of the environment on the WT is enormous. From the perspective of electricity cost, it can be found that when the environment is more complex, the electricity cost will also rise, but due to the optimization of the strategy, the trend of the equipment operation time changing with ρ is not obvious.

5.6.3. Rationality analysis of assembly cost rate

This section analyzes the rationality of the assembly cost rate. Fig. 3 can be regarded as showing that the number of freezing days per week follows a gaussian distribution. The above text simply summarizes the freezing cost rate as a staircase distribution function, with the intention of penalizing activities such as the transportation of equipment and personnel allocation during the freezing period. This section considers the weekly average freezing dates in the plateau environment to follow a normal distribution, thereby verifying the rationality of the step assembly cost rate distribution function.

Fig. 9 (a) presents the optimization results of the assembly cost rate distribution function used in this paper, which is a staircase distribution function. Fig. 9 (b) shows the assembly cost rate distribution function obtained by fitting the original data with a gaussian.

distribution. It can be observed from the figures that the solutions obtained using both the gaussian distribution and the staircase distribution functions are similar, with maintenance costs being \$3.05/MWh and operation duration being 299.95 months, which are basically consistent with the solutions in Section 5.4. This indicates that the staircase distribution function used in this paper is reasonable. At the same time, when using the gaussian distribution function, the initial convergence speed of the algorithm is relatively slow. The algorithm was implemented in MATLAB® using a computer equipped with a 12th Gen Intel Core CPU at 2.0 GHz and 64 GB of RAM. When using the gaussian distribution function, the running time was about 2 h, while when using the staircase distribution function, the running time was reduced to 1.5 h. This demonstrates that using a staircase distribution function to approximate the assembly cost rate distribution function is both reasonable and efficient.

5.6.4. The impact of technological advancement on maintenance

The technological iteration and commercialization of wind power generation are progressing very rapidly worldwide. To balance the technological iteration with ongoing maintenance, this section analyzes the impact of technological advancements on the maintenance process of WTs. With the progress of artificial intelligence technology, the fault diagnosis of gearboxes has evolved from initial manual identification to current intelligent diagnosis, which has not only improved diagnostic accuracy but also reduced the cost of diagnosis. Therefore, this paper primarily models the impact of technological iteration as a decrease in maintenance costs over time.

As technology advances over time, almost all costs associated with equipment maintenance will gradually decrease, such as preventive maintenance costs, corrective maintenance costs, regular inspection costs, and equipment purchase costs [78]. The reduction of these costs can essentially be seen as the depreciation of maintenance technology costs. With the rapid development of the wind power market, the cost of maintenance will become increasingly lower [16]. Therefore, we add a depreciation factor to the numerator of Eq. (19), and the new

Table	8				
Range	of nondominated	solution	strategies	for	different <i>p</i>

ρ	Maintenance cost(\$/MWh)	Run time(months)
0.05	2.892–2.916	285.7-289.5
0.1	2.837-2.974	284.1-288.6
0.15	2.923-3.006	288.6-292.5
0.2	2.952-3.105	286.3-289.8

maintenance cost function is shown as Eq. (21).

$$\min C = \frac{C_{all}R}{H \cdot E} \tag{21}$$

In the equation, $R = \frac{1}{(1+\delta)^{g(H)}}$, where R is the depreciation factor, and δ represents the depreciation rate. Since the technological iteration process is relatively long, it is set asg(H) = H/12to characterize the gradual progress of technology. The number 12 here indicates that technological iteration occurs once a year, but it can also be set to other appropriate values. However, the focus of this paper is not on how much this value should be set, but on the changes in maintenance strategies under the condition of technological progress. Fig. 10 shows the trend of changes in maintenance objectives as the depreciation factor varies.

The discount on maintenance costs due to technological progress is usually low, as a revolutionary change in an industry typically requires decades or even centuries of technological accumulation. In this paper, three discount rates are selected to explore the impact of technological progress on the economic viability of maintenance. From Fig. 10, we can see that as the discount rate increases, the cost of electricity from wind power gradually decreases, but the reduction is relatively small and tends to show a saturation trend. The results indicate that the usual means for wind power companies to improve economic viability is not to adopt more advanced maintenance technologies, as the effect on reducing maintenance costs is limited. Of course, if companies are willing to make appropriate investments in updating maintenance equipment or developing innovative maintenance technologies, they can still reduce some costs.

6. Conclusion

To solve the problem of low economic efficiency and more complex performance degradation of WTs in complex operating environments on high plateaus compared to flatlands, a maintenance strategy is proposed for plateau WTs that considers degradation states and time-varying maintenance costs. This paper models the life decay of each state of the equipment as a random process that follows an exponential distribution, establishing a model for the time-variant maintenance costs of WTs. A case study was conducted on a plateau wind farm in Guizhou, China, optimizing the maintenance cost per unit of electricity and operational lifespan, thereby determining the optimal level of maintenance resource input and maintenance frequency for each state.

The study demonstrates that compared to FFMS, our strategy can reduce the maintenance cost per unit of electricity by 24.07 %, while extending the operational lifespan of the equipment by 11.58 %. In contrast to FRRS, our strategy can reduce the maintenance cost per unit of electricity by 18.21 % and extend the operational lifespan of the equipment by 3.69 %. When making maintenance decisions for wind farms, the optimization strategy proposed in this paper eliminates the need to focus on when equipment failures will occur; instead, it requires deciding on the resources needed for each maintenance and the frequency of maintenance. Compared to opportunistic or state-based maintenance strategies, our approach requires only condition monitoring at any time after a WT is put into operation, followed by optimizing decisions on the ratio of maintenance resources and maintenance frequency. Furthermore, the optimization results indicate that the FRRS traditionally adopted by wind farms, where the maintenance resources are the same in the early and late stages of equipment operation, affects the operational condition of the equipment in the later stages. In the long term, during the maintenance of plateau WTs, a low level of resources but frequent maintenance is required in the early stages of operation, while a high level of resources and higher frequency of maintenance is needed in the mid-to-late stages.

This work primarily focuses on the maintenance costs and operational lifespan of the equipment, without considering the efficiency changes of WTs during operation and equipment depreciation. In



Fig. 9. Optimization results of different assembly cost rate distribution functions.



The Impact of Technological Advancement on Maintenance

Fig. 10. The impact of technological advancement on maintenance.

practice, improving the efficiency of WTs can better enhance the economic viability of wind power, while equipment depreciation affects the utility of the maintenance plan. Therefore, our future research can revolve around these key points, improving the model from a broader perspective to adapt to more complex situations. Additionally, conducting experiments in wind farms will also be a focus of our next phase of work.

Funding

This study is supported by National Natural Science Foundation of China (No. 71962004), National Natural Science Foundation of China (No. 72464005), and National Key Research and Development Program of China (No. 2022YFE0205300).

CRediT authorship contribution statement

Huakang Tang: Writing - original draft, Methodology, Data

curation, Conceptualization. **Honglei Wang:** Writing – review & editing, Visualization, Supervision, Investigation. **Chengjiang Li:** Writing – review & editing, Visualization, Formal analysis.

Declaration of competing interest

The authors declare that they have no known competing financial

Appendix A. Optimization results of operation at other times

interests or personal relationships that could have appeared to influence the work reported in this study.

Data availability

No data was used for the research described in the article.



Fig. A1. Optimization results when put into operation in April and May.



Fig. A2. Optimization results when put into operation in July and August.



Fig. A3. Optimization results from the operation launched in October.

References

- He G, Lin J, Sifuentes F, Liu X, Abhyankar N, Phadke A. Rapid cost decrease of renewables and storage accelerates the decarbonization of China's power system. Nat Commun 2020;11:2486.
- [2] Dai J, Yang X, Wen L. Development of wind power industry in China: a comprehensive assessment. Renew Sust Energ Rev 2018;97:156–64.
- [3] Elsner P. Continental-scale assessment of the African offshore wind energy potential: spatial analysis of an under-appreciated renewable energy resource. Renew Sust Energ Rev 2019;104:394–407.
- [4] Ohunakin OS, Matthew OJ, Adaramola MS, Atiba OE, Adelekan DS, Aluko OO, et al. Techno-economic assessment of offshore wind energy potential at selected sites in the Gulf of Guinea. Energy Convers Manag 2023;288:117110.
- [5] Gyatso N, Li Y, Gao Z, Wang Q, Li S, Yin Q, et al. Wind power performance assessment at high plateau region: a case study of the wind farm field test on the Qinghai-Tibet plateau. Appl Energy 2023;336:120789.
- [6] Dan Z, Che Y, Wang X, Zhou P, Han Z, Bu D, et al. Environmental, economic, and energy analysis of municipal solid waste incineration under anoxic environment in Tibet plateau. Environ Res 2023;216:114681.
- [7] Zhu Y, Zhong S, Shen L, Li D, Hou X. From potential to utilization: exploring the optimal layout with the technical path of wind resource development in Tibet. Energy Convers Manag 2024;304:118231.
- [8] Pfaffel S, Faulstich S, Rohrig K. Performance and reliability of wind turbines: A review energies 2017;10:1904.
- [9] Roga S, Dahiwale H, Bardhan S, Sinha S. Wind energy potential assessment: a case study in Central India. Proceedings of the Institution of Civil Engineers-Energy 2024:1–19.
- [10] Elgendi M, AlMallahi M, Abdelkhalig A, Selim MY. A review of wind turbines in complex terrain. International Journal of Thermofluids 2023;17:100289.
- [11] Tang W, Xu S, Zhou X, Yang K, Wang Y, Qin J, et al. Meeting China's electricity demand with renewable energy over Tibetan plateau. Sci Bull 2023;68:39–42.
- [12] Fan J-L, Huang X, Shi J, Li K, Cai J, Zhang X. Complementary potential of windsolar-hydro power in Chinese provinces: based on a high temporal resolution multiobjective optimization model. Renew Sust Energ Rev 2023;184:113566.
- [13] Zheng Y, Liu H, Chamorro LP, Zhao Z, Li Y, Zheng Y, et al. Impact of turbulence level on intermittent-like events in the wake of a model wind turbine. Renew Energy 2023;203:45–55.
- [14] Qiu L, He L, Lu H, Liang D. Systematic potential analysis on renewable energy centralized co-development at high altitude: a case study in Qinghai-Tibet plateau. Energy Convers Manag 2022;267:115879.
- [15] Abdeslame D, Merzouk NK, Mekhtoub S, Abbas M, Dehmas M. Estimation of power generation capacities of a wind farms installed in windy sites in Algerian high plateaus. Renew Energy 2017;103:630–40.
- [16] Kanyako F, Janajreh I. Implementation and economical study of HAWT under different wind scenarios. Sustain Cities Soc 2015;15:153–60.
- [17] Gökçek M, Genç MS. Evaluation of electricity generation and energy cost of wind energy conversion systems (WECSs) in Central Turkey. Appl Energy 2009;86: 2731–9.
- [18] Chehouri A, Younes R, Ilinca A, Perron J. Review of performance optimization techniques applied to wind turbines. Appl Energy 2015;142:361–88.
- [19] Njiri JG, Beganovic N, Do MH, Söffker D. Consideration of lifetime and fatigue load in wind turbine control. Renew Energy 2019;131:818–28.

- [20] Dao C, Kazemtabrizi B, Crabtree C. Wind turbine reliability data review and impacts on levelised cost of energy. Wind Energy 2019;22:1848–71.
- [21] Eryilmaz S, Navarro J. A decision theoretic framework for reliability-based optimal wind turbine selection. Reliab Eng Syst Saf 2022;221:108291.
- [22] Shafiee M, Sørensen JD. Maintenance optimization and inspection planning of wind energy assets: models, methods and strategies. Reliab Eng Syst Saf 2019;192: 105993.
- [23] Zhong S, Pantelous AA, Goh M, Zhou J. A reliability-and-cost-based fuzzy approach to optimize preventive maintenance scheduling for offshore wind farms. Mech Syst Signal Process 2019;124:643–63.
- [24] Perez-Canto S, Rubio-Romero JC. A model for the preventive maintenance scheduling of power plants including wind farms. Reliab Eng Syst Saf 2013;119: 67–75.
- [25] Nielsen JJ, Sørensen JD. On risk-based operation and maintenance of offshore wind turbine components. Reliab Eng Syst Saf 2011;96:218–29.
- [26] Stoyanov D, Nixon J, Sarlak H. Analysis of derating and anti-icing strategies for wind turbines in cold climates. Appl Energy 2021;288:116610.
- [27] Swenson L, Gao L, Hong J, Shen L. An efficacious model for predicting icinginduced energy loss for wind turbines. Appl Energy 2022;305:117809.
- [28] Zhang Z, Yang L. State-based opportunistic maintenance with multifunctional maintenance windows. IEEE Trans Reliab 2020;70:1481–94.
- [29] de Jonge B, Teunter R, Tinga T. The influence of practical factors on the benefits of condition-based maintenance over time-based maintenance. Reliab Eng Syst Saf 2017;158:21–30.
- [30] Radünz WC, Sakagami Y, Haas R, Petry AP, Passos JC, Miqueletti M, et al. Influence of atmospheric stability on wind farm performance in complex terrain. Appl Energy 2021;282:116149.
- [31] Barlow R, Hunter L. Optimum preventive maintenance policies. Oper Res 1960;8: 90–100.
- [32] Shin W, Han J, Rhee W. AI-assistance for predictive maintenance of renewable energy systems. Energy 2021;221:119775.
- [33] Havinga MJ, de Jonge B. Condition-based maintenance in the cyclic patrolling repairman problem. Int J Prod Econ 2020;222:107497.
- [34] Lin Z, Cevasco D, Collu M. A methodology to develop reduced-order models to support the operation and maintenance of offshore wind turbines. Appl Energy 2020;259:114228.
- [35] Zitrou A, Bedford T, Walls L. A model for availability growth with application to new generation offshore wind farms. Reliab Eng Syst Saf 2016;152:83–94.
- [36] Perveen R, Kishor N, Mohanty SR. Off-shore wind farm development: present status and challenges. Renew Sust Energ Rev 2014;29:780–92.
- [37] Kraudzun T. Bottom-up and top-down dynamics of the energy transformation in the eastern Pamirs of Tajikistan's Gorno Badakhshan region. Central Asian Survey 2014;33:550–65.
- [38] Song D, Liu J, Yang J, Su M, Yang S, Yang X, et al. Multi-objective energy-cost design optimization for the variable-speed wind turbine at high-altitude sites. Energy Convers Manag 2019;196:513–24.
- [39] Staffell I, Pfenninger S. Using bias-corrected reanalysis to simulate current and future wind power output. Energy 2016;114:1224–39.
- [40] Schouten TN, Dekker R, Hekimoğlu M, Eruguz AS. Maintenance optimization for a single wind turbine component under time-varying costs. Eur J Oper Res 2022;300: 979–91.
- [41] Jasiūnas J, Heikkinen T, Lund PD, Láng-Ritter I. Resilience of electric grid to extreme wind: considering local details at national scale. Reliab Eng Syst Saf 2023; 232:109070.

H. Tang et al.

Applied Energy 377 (2025) 124464

- [42] Scheu MN, Kolios A, Fischer T, Brennan F. Influence of statistical uncertainty of component reliability estimations on offshore wind farm availability. Reliab Eng Syst Saf 2017;168:28–39.
- [43] Ghamlouch H, Fouladirad M, Grall A. The use of real option in condition-based maintenance scheduling for wind turbines with production and deterioration uncertainties. Reliab Eng Syst Saf 2019;188:614–23.
- [44] Igba J, Alemzadeh K, Henningsen K, Durugbo C. Effect of preventive maintenance intervals on reliability and maintenance costs of wind turbine gearboxes. Wind Energy 2015;18:2013–24.
- [45] Bhardwaj U, Teixeira A, Soares CG. Reliability prediction of an offshore wind turbine gearbox. Renew Energy 2019;141:693–706.
- [46] Yürüşen NY, Rowley PN, Watson SJ, Melero JJ. Automated wind turbine maintenance scheduling. Reliab Eng Syst Saf 2020;200:106965.
- [47] Wang J, Liang Y, Zheng Y, Gao RX, Zhang F. An integrated fault diagnosis and prognosis approach for predictive maintenance of wind turbine bearing with limited samples. Renew Energy 2020;145:642–50.
- [48] Barlow E, Bedford T, Revie M, Tan J, Walls L. A performance-centred approach to optimising maintenance of complex systems. Eur J Oper Res 2021;292:579–95.
- [49] Famoso F, Brusca S, D'Urso D, Galvagno A, Chiacchio F. A novel hybrid model for the estimation of energy conversion in a wind farm combining wake effects and stochastic dependability. Appl Energy 2020;280:115967.
- [50] Eti MC, Ogaji S, Probert S. Reducing the cost of preventive maintenance (PM) through adopting a proactive reliability-focused culture. Appl Energy 2006;83: 1235–48.
- [51] Rasmekomen N, Parlikad AK. Maintenance optimization for asset systems with dependent performance degradation. IEEE Trans Reliab 2013;62:362–7.
- [52] Ajukumar V, Gandhi O. Evaluation of green maintenance initiatives in design and development of mechanical systems using an integrated approach. J Clean Prod 2013;51:34–46.
- [53] Besnard F, Bertling L. An approach for condition-based maintenance optimization applied to wind turbine blades. IEEE Transactions on Sustainable Energy 2010;1: 77–83.
- [54] Pookkuttath S, Rajesh Elara M, Sivanantham V, Ramalingam B. AI-enabled predictive maintenance framework for autonomous mobile cleaning robots. Sensors 2021;22:13.
- [55] Nouri Gharahasanlou A, Ataei M, Khalokakaie R, Barabadi A, Einian V. Risk based maintenance strategy: a quantitative approach based on time-to-failure model. Int J Syst Assur Eng Manag 2017;8:602–11.
- [56] Nourelfath M, Nahas N, Ben-Daya M. Integrated preventive maintenance and production decisions for imperfect processes. Reliab Eng Syst Saf 2016;148:21–31.
 [57] Camci F. System maintenance scheduling with prognostics information using
- [57] Camel F. System maintenance scheduling with prognostics information using genetic algorithm. IEEE Trans Reliab 2009;58:539–52.
- [58] De Jonge B, Scarf PA. A review on maintenance optimization. Eur J Oper Res 2020; 285:805–24.
- [59] Vatn J, Hokstad P, Bodsberg L. An overall model for maintenance optimization. Reliab Eng Syst Saf 1996;51:241–57.

- [60] Papadopoulos P, Coit DW, Ezzat AA. Seizing opportunity: maintenance optimization in offshore wind farms considering accessibility, production, and crew dispatch. IEEE Transactions on Sustainable Energy 2021;13:111–21.
- [61] Lee D, Pan R. Predictive maintenance of complex system with multi-level reliability structure. Int J Prod Res 2017;55:4785–801.
- [62] Sanoubar S, Maillart LM, Prokopyev OA. Age-replacement policies under agedependent replacement costs. IISE Trans 2021;53:425–36.
- [63] Van Noortwijk JM. A survey of the application of gamma processes in maintenance. Reliab Eng Syst Saf 2009;94:2–21.
- [64] Liu B, Pang J, Yang H, Zhao Y. Optimal condition-based maintenance policy for leased equipment considering hybrid preventive maintenance and periodic inspection. Reliab Eng Syst Saf 2024;242:109724.
- [65] Yuanru P, Qinming L, Wenyuan L, Chunming Y. A research on periodic multimaintenance strategy of leased equipment based on failure state evolution. Ind Eng J 2018;21:57.
- [66] Wen X, Xu Z. Wind turbine fault diagnosis based on ReliefF-PCA and DNN. Expert Syst Appl 2021;178:115016.
- [67] Do P, Voisin A, Levrat E, Iung B. A proactive condition-based maintenance strategy with both perfect and imperfect maintenance actions. Reliab Eng Syst Saf 2015; 133:22–32.
- [68] Tin T, Sovacool BK, Blake D, Magill P, El Naggar S, Lidstrom S, et al. Energy efficiency and renewable energy under extreme conditions: case studies from Antarctica. Renew Energy 2010;35:1715–23.
- [69] Hu JW, Chen P. Predictive maintenance of systems subject to hard failure based on proportional hazards model. Reliab Eng Syst Saf 2020;196.
- [70] Yang L, Ma XB, Zhao Y. A condition-based maintenance model for a three-state system subject to degradation and environmental shocks. Comput Ind Eng 2017; 105:210–22.
- [71] Wang W. Analysis of fault detection in rolling element bearings. IEEE Instrum Meas Mag 2021;24:42–9.
- [72] de Jong P, Dargaville R, Silver J, Utembe S, Kiperstok A, Torres EA. Forecasting high proportions of wind energy supplying the Brazilian northeast electricity grid. Appl Energy 2017;195:538–55.
- [73] Li QS, Zhang Y, Cheng F, Yao LZ, Xu YY, Liang S, et al. Generation expansion planning for Guizhou province based on the complementary characteristics of wind and solar. Energy Rep 2022;8:574–84.
- [74] Cai DP, Tao LF, Yang XQ, Sang XZ, Fang JB, Sun XG, et al. A climate perspective of the quasi-stationary front in southwestern China: structure, variation and impact. Clim Dyn 2022;59:547–60.
- [75] Faulstich S, Hahn B, Tavner PJ. Wind turbine downtime and its importance for offshore deployment. Wind Energy 2011;14:327–37.
- [76] Kennedy J, Eberhart R. Particle swarm optimization. Proceedings of ICNN'95international conference on neural networks. ieee; 1995. p. 8–1942.
- [77] Deb K, Pratap A, Agarwal S, Meyarivan T. A fast and elitist multiobjective genetic algorithm: NSGA-II. IEEE Trans Evol Comput 2002;6:182–97.
- [78] Shrimali G, Pusarla S, Trivedi S. Did accelerated depreciation result in lower generation efficiencies for wind plants in India: an empirical analysis. Energy Policy 2017;102:154–63.